Author's response to reviews

Title: Hybrid management of a spontaneous ileo-iliac arteriovenous fistula.

Authors:

Gavin C O'Brien (gavinobrien@rcsi.ie)
Colm Murphy (colmmurphy1978@gmail.com)
Zenia Martin (zmartin@rcsi.ie)
Naseem Haider (talk2naseem@eircom.net)
Mary P Colgan (mpcolgan@stjames.ie)
Prakash Madhavan (pmadhavan@stjames.ie)
Dermot Moore (dmoore@stjames.ie)
Sean M O'Neill (smoneill@stjames.ie)

Version: 3 Date: 13 June 2011

Author's response to reviews:

I resubmitted this paper on May 27th but received an email from Ivan Serraon on June 7th requesting some further changes - these are now included:
Arrows have been added to the figures.
":a case report" has been added to the title.
The patients ethnicity is in both the abstract and main text.
The abstract's subheadings have been edited.
An introduction has been included.
'No competing interest' is declared on the title page.
An 'authors contribution' subheading is included.